Dear reviewers,

Thank you for your time invested in reviewing presented manuscript. We are pleased that you found our manuscript interesting and worthy for publication. We have made suggested changes along with the English language editing.

Please find our replies to your comments below.

Kind regards,

Anna Mrzljak, MD, PhD

Reviewer #1:
Specific Comments to Authors: This study aimed to determine the parvovirus B19 serological status and DNAemia in kidney, liver and pancreas transplant candidates. This a very important subject, due the not fully elucidated and understimated clinical implications of parvovirus infection in transplanted patients. The main limitation of the study is not reporting the incidence of clinical parvovirus infection in the postoperative period of these patients, reflecting more accurately the real significance of positive ELISA tests detected in the preoperative period. But, this manuscript is highly important for this debate, and the lack of these results could be partially healed by a more extensive discussion of the updated bibliography on the subject (discussion session). Congratulations for the authors.

To answer the point raised by the reviewer #1, we broadened the discussion with possible implications in post-transplant management and further plans for DNA and antibody monitoring. We also performed language polishing of the manuscript, and we hope that the revised manuscript will meet the publication requirement.

Reviewer #2:
Specific Comments to Authors: This study determined the serological status and DNAemia of parvovirus B19 in 131 renal, liver and pancreatic transplant candidates, and prospectively speculated that pre-transplant serological studies could simplify the diagnosis of post-transplant anemia, with novel themes and good research significance. The study subjects were the pre-transplant patients. So how to choose transplant type? What are the differences in the clinical characteristics of these four types of patients?

To answer the question of reviewer #2, we elaborated the differences in the clinical characteristics of these four types of patients. We also performed language polishing of the manuscript, and we hope that the revised manuscript will meet the publication requirement.